Author's response to reviews

Title: Rapid improvement in verbal fluency and aphasia following perispinal etanercept in Alzheimer's disease

Authors:

Edward L. Tobinick (etmd@ucla.edu)
Hyman Gross (hymangross@yahoo.com)

Version: 6 Date: 30 June 2008

Author's response to reviews:

The title was shortened for improved readability.
The abstract was redone to meet the form requirements of the BMC Medical journals.
The manuscript is unchanged.